\begin{table}%t2
\caption{\label{tab.RCB.GenInfo}General information on the new confirmed and candidate EROS-2 Magellanic RCB and DYPer~stars.} 
%%\centerline

 {\small \begin{tabular}{llll}
\hline \hline
EROS-2 Magellanic & EROS-2 star & Coordinates ($J_{2000}$) & Other identifier  \\
RCB name & identifier &  & \\
\hline
 &  &  & \\
 &  &  New confirmed RCB stars &  \\
\object{EROS2-LMC-RCB-1}$^{S2}$  &  \object{lm0094n12459}  &  05:14:40.17 --69:58:40.06  & [KDM2001]-2831 ; MACHO-5.5489.623\\
\object{EROS2-LMC-RCB-2}$^{S2}$  &  \object{lm0105m6306}  &  05:10:28.50 --69:47:04.54  & [KDM2001]-2373 ; SP77-39-12$^a$ ; MACHO-5.4887.14 \\
\object{EROS2-LMC-RCB-3}$^{S1}$  &  \object{lm0174k9881}  &  04:59:35.78 --68:24:44.68  & [KDM2001]-1296 ; SP77-30-16$^a$ \\
\object{EROS2-LMC-RCB-4}$^{S6+S7}$  &  \object{lm0720k11917}  &  05:39:36.97 -71:55:46.42  & MACHO-27.9574.93 \\
\object{EROS2-LMC-RCB-5}$^{S1}$  &  \object{lm0745l21924}  &  06:04:05.42 --72:51:22.73  & [KDM2001]-7101 ; SP77-65-2$^a$ \\
\object{EROS2-LMC-RCB-6}  &  \object{lm0864m16909}  &  06:12:10.48 --74:05:10.16  & \\
\hline
 &  &  & \\
 &  & New confirmed DYPer stars &  \\
\object{EROS2-LMC-DYPer-1}$^{S2}$ & \object{lm0112l16201} & 04:56:28.47 --69:39:12.57 & [KDM2001]-1033 ; MACHO-17.2590.220\\
\object{EROS2-LMC-DYPer-2}$^{S1}$ & \object{lm0405m10336} & 06:18:46.59 --67:00:59.59 & [KDM2001]-7544 \\
\object{EROS2-LMC-DYPer-3}$^{S1}$ & \object{lm0485n10275} & 05:44:04.05 --65:46:00.84 & [KDM2001]-5900 \\
\object{EROS2-LMC-DYPer-4}$^{S1}$ & \object{lm0013n33996} & 05:25:54.95 --69:43:54.03 & MACHO-77.7429.64 ; OGLE2-LMC-SC4-450098\\
\object{EROS2-LMC-DYPer-5}$^{S1}$ &  \object{lm0730k16380}  &  05:50:15.47 --71:57:39.66  & [KDM2001]-6448 ; MACHO-51.11267.9 \\
\object{EROS2-LMC-DYPer-6}$^{S4}$ &  \object{lm0020n28519} & 05:29:27.19 --69:22:32.61 & MACHO-82.7918.2694 \\
\object{EROS2-SMC-DYPer-3}$^{S2}$ &  \object{sm0067m2668}  &  00:55:54.97 --72:35:12.27  & RAW-961$^e$ ; OGLE2-SMC-SC7-368043 \\
\hline
 &  &  & \\
 &  &  New RCB candidates &  \\
\object{EROS2-LMC-RCB-7} & \object{lm0577n19651} & 05:24:43.68 --71:47:26.40 & MACHO-21.7156.850; MSX-428$^d$ \\
\object{EROS2-LMC-RCB-8} & \object{lm0675k8409} & 04:58:48.37 --72:37:43.37 & MSX-184$^d$ \\
\object{EROS2-SMC-RCB-4} & \object{sm0053n9475} & 01:04:52.89 --72:04:02.64  & OGLE2-SMC SC10-107856 \\
\hline
 &  &  & \\
 &  &  New DYPer candidates &  \\
\object{EROS2-LMC-DYPer-7}$^{S1}$ & \object{lm0456m10958} & 05:19:48.13 --65:58:22.55 & [KDM2001]-3340 ; MACHO-63.6396.1154\\
\object{EROS2-LMC-DYPer-8}$^{S1}$ & \object{lm0593k30837} & 05:39:00.21 --70:58:01.25  & [KDM2001]-5419 ; MACHO-14.9467.9 \\
\object{EROS2-LMC-DYPer-9}$^{S1}$ & \object{lm0121k19954}  &  04:52:58.35 --69:10:27.42  & [KDM2001]-804 ; LMC-BM 5-9$^b$ \\
\object{EROS2-LMC-DYPer-10}$^{S1}$ & \object{lm0257n7946}  &  06:05:35.32 --68:54:54.05  & [KDM2001]-7151 \\
\object{EROS2-LMC-DYPer-11}$^{S7}$ & \object{lm0026n10671} &  05:29:22.61  --70:19:35.65  & MACHO-7.7904.11 ;  [KDM2001]-4359\\
\object{EROS2-LMC-DYPer-12}$^{S7}$ & \object{lm0335m9403} & 05:29:26.21  --67:00:22.64  & MACHO-60.7954.68 ; [KDM2001]-4370\\
\object{EROS2-LMC-DYPer-13}$^{S4}$ & \object{lm0010k3834} & 05:20:22.92 --69:04:43.11 & MACHO-80.6471.2523 \\
\object{EROS2-LMC-DYPer-14}$^{S4}$ & \object{lm0020k5163} & 05:27:32.33 --69:05:35.48 & \\
\object{EROS2-LMC-DYPer-15}$^{S5}$ & \object{lm0022n15676} & 05:30:43.25 --69:38:51.84 & MACHO-77.8156.16 ; BSDL2075$^c$\\
\object{EROS2-LMC-DYPer-16}$^{S4}$ & \object{lm0034l8100} & 05:35:15.49 --69:57:43.84 & MACHO-81.8877.24 \\
\object{EROS2-LMC-DYPer-17}$^{S5}$ & \object{lm0341m22605} & 05:35:24.08 --66:24:15.17 & [KDM2001]-5039\\
\object{EROS2-SMC-DYPer-4}$^{S2}$ & \object{sm0061m7643} & 00:56:35.47 --71:32:32.66 & [MH95]-672 \\
\object{EROS2-SMC-DYPer-5}$^{S8}$ & \object{sm0010l4971} & 00:47:41.71 --73:06:16.38 & MACHO-212.15793.25 ; RAW-421 \\
\object{EROS2-SMC-DYPer-6}$^{S8}$ & \object{sm0101n8018} & 00:44:56.40 --73:12:25.02 & MACHO-212.15621.153 \\
\hline
\end{tabular}}
\medskip
$^a$ \citet{1979ApJ...230..724R}; $^b$~\citet{1990AJ....100..674B}; $^c$~\citet{1999AJ....117..238B}; $^d$~\citet{2001AJ....122.1844E}; $^e$~\citet{1993AetAS...97..603R}. Spectra epochs (JD-2450000): $^{S1} = 4510.0$, $^{S2}= 4511.0$, $^{S3}= 4645.0$, $^{S4} = 4646.0$, $^{S5} = 4647.0$, $^{S6} = 4747.0$, $^{S7} = 4820.0$, $^{S8} = 4821.0$.
\end{table}